Author's response to reviews

Title: Clinical action measures improve the reliability of feedback on quality of care in diabetes centres: a retrospective cohort study.

Authors:

Astrid Lavens (astrid.lavens@wiv-isp.be)
Kris Doggen (kris.doggen@wiv-isp.be)
Chantal Mathieu (chantal.mathieu@uz.kuleuven.be)
Frank Nobels (frank.nobels@olvz-aalst.be)
Evy Vandemeulebroucke (evy.vandemeulebroucke@azjanportaels.be)
Michel Vandenbroucke (michel.vandenbroucke@emmaus.be)
Ann Verhaegen (Ann.Verhaegen@zna.be)
Viviane Van Casteren (viviane.vancasteren@wiv-isp.be)

Version: 4  Date: 4 August 2016

Author's response to reviews: see over
Dear Editor,

I’m pleased to submit the revision of my manuscript.

Please find below a detailed description of the editorial revisions made.

Sincerely,

Astrid Lavens
Editorial revisions made:

1. Please clarify in the manuscript if the IQED data are freely available or if permission was required to access them, and if so, from whom

The IQED project is a project that is managed by our institute. As project manager of the IQED project, I have access to the data. Those data are not public available. See line 104 – 107
See also declaration “availability of data and materials”.

2. Please specifically mention the contribution of each author by their initials in the Authors Contributions section

See line 438 - 441

3. Include titles/legends for the additional files at the end of the main manuscript and ensure that these files are referred to in the main text

See line 614 - 618

4. In accordance with BioMed Central editorial policies and our revised formatting guidelines, all submissions to BMC journals must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal’s Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article).

List of abbreviations: See line 410 - 423

Declarations: see line 425 - 443